Author’s response to reviews

Title: Change in Forced Vital Capacity and Associated Subsequent Outcomes in Patients with Newly Diagnosed Idiopathic Pulmonary Fibrosis

Authors:

William Reichmann (bill.reichmann@gmail.com)

Yanni Yu (Yanni.Yu@boehringer-ingleheim.com)

Dendy Macaulay
(Dendy.Macaulay@analysisgroup.com; Todor.Totev@analysisgroup.com; Emi.Terasawa@analysisgroup.com)

Eric Wu (Eric.Wu@analysisgroup.com)

Steven Nathan (Steven.Nathan@inova.org)

Version: 2 Date: 10 Dec 2015

Author’s response to reviews:

Editorial Requests 1-3 (1. Remove the appendices from the main manuscript and upload them as supplemental files; 2. Include Abbreviation section after the conclusion; and 3. Move the information regarding funding from the Acknowledgments to the Competing Interests) have been addressed in the manuscript.